Author's response to reviews

Title: Applying psychological theories to evidence-based clinical practice: Identifying factors predictive of managing upper respiratory tract infections without antibiotics.

Authors:

Martin P Eccles (martin.eccles@ncl.ac.uk)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Marie Johnston (m.johnston@abdn.ac.uk)
Nick Steen (nick.steen@ncl.ac.uk)
Nigel B Pitts (n.b.pitts@chs.dundee.ac.uk)
Ruth Thomas (r.e.thomas@abdn.ac.uk)
Elizabeth Glidewell (e.glidewell@abdn.ac.uk)
Graeme MacLennan (g.maclennnan@abdn.ac.uk)
Debbie Bonetti (d.bonetti@chs.dundee.ac.uk)
Anne Walker (Anne_walker@yahoo.co.uk)

Version: 5 Date: 28 July 2007

Author's response to reviews:

I have dealt with all of the revisions as requested in the production team's email.